Author's response to reviews

Title: A complex regional intervention to implement advance care planning in one town's nursing homes: a controlled inter-regional study. Study design

Authors:

Jürgen in der Schmitten (juergen.inderschmitten@med.uni-duesseldorf.de)
Sonja Rothärmel (sonja.rothaermel@jura.uni-augsburg.de)
Christine Mellert (christine.mellert@med.uni-duesseldorf.de)
Stephan Rixen (stephan.rixen@uni-bayreuth.de)
Bernard J Hammes (bjhammes@gundluth.org)
Linda Briggs (labriggs@gundluth.org)
Karl Wegscheider (k.wegscheider@uke.uni-hamburg.de)
Georg Marckmann (marckmann@lmu.de)

Version: 2 Date: 17 January 2011

Author's response to reviews:

Dear editorial team,

we made the one major and the several minor revisions according the instructions in your e-mail of 01/13/11. Hopefully, everything is as required!

Please let us know if we can do anything else.

Thank you very much for the incredibly fast processing of our manuscript - that's a great service!!

Best regards

Jürgen in der Schmitten, Georg Marckmann